Author's response to reviews

Title: Early Life Environment and Natural History of Inflammatory Bowel Diseases

Authors:

Abra Guo (aguo1@mgh.harvard.edu)
Betsy Stevens (bstevens1@partners.org)
Robin Wilson (rgwilson@mgh.harvard.edu)
Caitlin Russell (cnrussell@mgh.harvard.edu)
Melissa Cohen (mcohen4@mgh.harvard.edu)
Holly Sturgeon (hsturgeon@mgh.harvard.edu)
Anna Thornton (athornton4@mgh.harvard.edu)
Cosmas Giallourakis (cgiallourakis@mgh.harvard.edu)
Hamed Khalili (hkhalili@mgh.harvard.edu)
Deanna Nguyen (deanna.nguyen@mgh.harvard.edu)
Jenny Sauk (jsauk@partners.org)
Vijay Yajnik (vyajnik@partners.org)
Ramnik J Xavier (rxavier@mgh.harvard.edu)
Ashwin N Ananthakrishnan (aananthakrishnan@partners.org)

Version: 6 Date: 26 November 2014

Author's response to reviews: see over
Editorial Requirements:

1. Competing interests

A competing interest exists when your interpretation of data or presentation of information may be influenced by your personal or financial relationship with other people or organizations. Authors must disclose any financial competing interests; they should also reveal any non-financial competing interests that may cause them embarrassment were they to become public after the publication of the manuscript.

Authors are required to complete a declaration of competing interests. All competing interests that are declared will be listed at the end of published articles. Where an author gives no competing interests, the listing will read 'The author(s) declare that they have no competing interests'.

When completing your declaration, please consider the following questions:

Financial competing interests

In the past three years have you received reimbursements, fees, funding, or salary from an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? Is such an organization financing this manuscript (including the article-processing charge)? If so, please specify.
Do you hold any stocks or shares in an organization that may in any way gain or lose financially from the publication of this manuscript, either now or in the future? If so, please specify.
Do you hold or are you currently applying for any patents relating to the content of the manuscript? Have you received reimbursements, fees, funding, or salary from an organization that holds or has applied for patents relating to the content of the manuscript? If so, please specify.
Do you have any other financial competing interests? If so, please specify.

Non-financial competing interests

Are there any non-financial competing interests (political, personal, religious, ideological, academic, intellectual, commercial or any other) to declare in relation to this manuscript? If so, please specify.
If you are unsure as to whether you, or one your co-authors, has a competing interest please discuss it with the editorial office.

Response: We have completed the competing interest information and presented it as requested in the acknowledgement section (Line 286).

2. Please include a description of the additional files in the MS.

Response: The additional file in the manuscript is the appendix with the questions. We have already provided a reference to this in the manuscript (line 109) and as a legend (Line 443).